Author's response to reviews

Title: A randomized controlled trial of acceptance and commitment therapy (ACT) for psychosis: study protocol

Authors:

Neil Thomas (neilthomas@swin.edu.au)
Frances Shawyer (frances.shawyer@monash.edu)
David Castle (david.castle@svhm.org.au)
David Copolov (david.copolov@monash.edu)
Steven Hayes (hayes@unr.edu)
John Farhall (j.farhall@latrobe.edu.au)

Version: 2
Date: 12 June 2014

Author's response to reviews: see over
12 June 2014

Carlo Rye Chua
Journal Editorial Office
BioMed Central
Floor 6, 236 Gray's Inn Road
London, WC1X 8HL

Dear Dr Chua,

Re: MS: 1709505463131700
Study Protocol
A randomized controlled trial of acceptance and commitment therapy (ACT) for psychosis: study protocol.

Thank you for your email dated 9 June 2014 regarding the above-named manuscript. I am writing to respond to the requests outlined in your email.

1. Ethical and Funding Approval Documentation

All ethics and funding approval documents have been sent to BMCSeriesEditorial@biomedcentral.com as requested, in a WinZip folder labelled, “Ethics and funding documentation_1709505463131700”. Note that there are actually ten ethics committees overseeing the project, not nine as stated in my initial cover letter.

2. Date of Trial Registration

18 April 2008

3. Requesting Ethics Statement

The manuscript has been amended to include the requested ethics statement and uploaded as “Thomas_protocol_revised”.

We will be pleased to provide you with any further information or clarifications that you may require.

Dr Frances Shawyer